Author's response to reviews

Title: Understanding discrepancies in parent-child reporting of emotional and behavioural problems: Effects of relational and socio-demographic factors

Authors:

Betty Van Roy (r.e.van@medisin.uio.no)
Berit Groholt (berit.groholt@medisin.uio.no)
Sonja Heyerdahl (sonja.heyerdahl@r-bup.no)
Jocelyne Clench-Aas (jocelyne.clench-aas@fhi.no)

Version: 4 Date: 26 June 2010

Author's response to reviews:

Editor-in -Chief
BMC Psychiatry

According to your request the following changes have been made in the manuscript:

Major revision:
Table 3 was cited in the measures section (p.7). This is now omitted as it is not essential in this section.

In this way it was not necessary to change the numerical order of the tables. It is preferable to present the tables in a numerical order according to the overall structure of the result section.

Minor revisions:
Keywords are removed from the abstract
All table placement text are removed

I apologize that I didn’t succeed in removing the visible vertical lines from the tables. I appreciate if you can make these changes for me, as you have suggested.

Yours sincerely

Betty Van Roy